Author’s response to reviews

Title: Video-assisted Radiofrequency Ablation for Pleural Disseminated Non-small Cell Lung Cancer

Authors:

Yaxing Shen (Shen.yaxing@zs-hospital.sh.cn)
Ming Zhong (zhong.ming@zs-hospital.sh.cn)
Wei Jiang (lung2792@163.com)
Hong Fan (Fan.hong@zs-hospital.sh.cn)
Hao Wang (Wang.hao@zs-hospital.sh.cn)
Qun Wang (doctorwangqun@yahoo.com.cn)

Version: 2 Date: 19 March 2013

Author’s response to reviews:

Dear Reviewer,

Many thanks for your comments, which have proven extremely valuable to us in improving the manuscript. We have made the following revisions according to your suggestions. The revised part was shown in black font in the manuscript. Thank you again for your kindly review.

Wei Jiang